Author’s response to reviews

Title: The SAIL databank: linking multiple health and social care datasets

Authors:

Ronan A Lyons (r.a.lyons@swansea.ac.uk)
Kerina H Jones (k.h.jones@swansea.ac.uk)
Gareth John (gareth.john@hsw.wales.nhs.uk)
Caroline J Brooks (c.brooks@swansea.ac.uk)
Jean-Philippe Verplancke (p.verplancke@swansea.ac.uk)
David V Ford (d.v.ford@swansea.ac.uk)
Ginevra Brown (g.brown@swansea.ac.uk)
Ken Leake (ken.leake@swansea.ac.uk)

Version: 4 Date: 15 January 2009

Author’s response to reviews:

Health Information Research Unit (HIRU)
Centre for Health Information Research & Evaluation (CHIRAL)
School of Medicine
Swansea University
Grove Building
CHIRAL suite
Singleton Park
Swansea
SA2 8PP

15th January 2009

BMC Medical Informatics & Decision Making

Dear Editor,

Re: Submission of revised paper (formatting changes) – The SAIL Databank: linking multiple health and social care datasets

Please find attached the revised version of the above-named paper with the formatting changes as requested.

The author details have been corrected: one of the authors was previously
named in a different way in the paper compared to the on-line version. The abbreviations have been formatted into a paragraph, not a list. The table title and legends have been re-positioned. The tables have been autoformatted using MS Word Simple1, and also have been converted to portrait layout. No vertical lines have been used, but horizontal lines have. Very minor typographical changes have been made to the abstract and so a new version has been uploaded.

I hope that the changes we have made will make the manuscript ready for publication. Please let us know if there are any other amendments to be made.

Thank you.

Yours sincerely,

Dr Kerina Jones
Research Development Fellow
HIRU

Tel: 01792 602764
Email: k.h.jones@swansea.ac.uk